Author's response to reviews

**Title:** Higher levels of emotional distress are associated with a higher risk of incident diabetes during 18 year follow-up: results from the British Household Panel Survey.

**Authors:**

Paula M.C. Mommersteeg (P.M.C.Mommersteeg@uvt.nl)
Raphael Herr (Raphael.Herr@medma.uni-heidelberg.de)
Wobbe P. Zijlstra (W.P.Zijlstra@uvt.nl)
Sven Schneider (sven.schneider@medma.uni-heidelberg.de)
François Pouwer (F.Pouwer@uvt.nl)

**Version:** 3  **Date:** 24 July 2012

**Author’s response to reviews:**

I have adapted the ethical approval statement:

(page 4) “Data collection was done in accordance to the declaration of Helsinki, and the study was ethically approved by the University of Essex, Institute for Social and Economic Research [10].”